Author's response to reviews

Title: AB0 blood group and prognosis in patients with pancreatic cancer

Authors:

Nuh N Rahbari (nuh.rahbari@med.uni-heidelberg.de)
Ulrich Bork (ulrich.bork@med.uni-heidelberg.de)
Ulf Hinz (ulf.hinz@med.uni-heidelberg.de)
Albrecht Leo (Albrecht.Leo@iktz-hd.de)
Johanna Kirchberg (johanna.kirchberg@med.uni-heidelberg.de)
Moritz Koch (moritz.koch@med.uni-heidelberg.de)
Markus W. Buchler (markus.buechler@med.uni-heidelberg.de)
Juergen Weitz (juergen.weitz@med.uni-heidelberg.de)

Version: 2  Date: 29 February 2012

Author's response to reviews:

Dear Editors

we made the following revisions to the manuscript according to the recommendations made by the editorial office:

1. We included a competing interests section at the end of the manuscript, before the reference list.

2. We included an Authors’ Contributions section at the end of the manuscript, before the reference list.

Revised passages have been highlighted yellow.

Kind regards

Nuh Rahbari